ORIGINAL REPORT

COMPARISON OF SHORT- AND MID-TERM OUTCOMES OF ITALIAN- AND GERMAN-SPEAKING PATIENTS AFTER AN INTERDISCIPLINARY PAIN MANAGEMENT PROGRAMME IN SWITZERLAND: A PROSPECTIVE COHORT STUDY

Thomas BENZ, MSc^{1,2}*, Susanne LEHMANN, RN¹, Roberto BRIOSCHI, MSc¹, Achim ELFERING, PhD², André AESCHLIMANN, MD¹ and Felix ANGST, MD, MPH¹ From the ¹Research Department, Rehabilitation Clinic "RehaClinic", Bad Zurzach, and ²Institute of Psychology, University of Bern,

From the "Research Department, Renabilitation Clinic "RenaClinic", Bad Zurzach, and "Institute of Psychology, University of Bern, Bern, Switzerland

Objective: To quantify and compare the course of health-related quality of life of immigrant native Italian-speaking and German-speaking patients before and after an interdisciplinary pain programme.

Design: Prospective cohort study with 1–12 month follow-up.

Subjects: Fibromyalgia, generalized widespread pain, and chronic non-specific back pain patients (Italian-speaking *n* = 96, German-speaking *n* = 199). Methods: Score changes measured with the Short Form 36 (SF-36) were compared with multivariate analysis using standardized mean differences (SMD), adjusted for sex, education and the baseline score. Results: At baseline, health of the Italian-speaking patients was worse than for the German-speaking patients. Adjusted SMDs showed significantly better improvements in the German group compared with the Italian group: SF-36 Physical functioning SMD = 0.54 (at discharge) and 0.49 (at 12 months), General health SMD = 0.71 and 0.44, Vitality SMD = 0.43 and 0.48 in one sample. In the other sample, the corresponding SMDs were 0.06 (discharge), 0.50 (3 months) and 0.47 (6 months) for Bodily pain. Conclusion: State of health was better and health improvements were greater in German-speaking patients compared with Italian-speaking patients. Patients with a migration background may have special needs in therapeutic management, and addressing these might enhance the positive outcome in the short- and mid-term.

Key words: rehabilitation; interdisciplinary pain management programme; fibromyalgia; back pain; language; socio-cultural factors.

Accepted Dec 7, 2018; Epub ahead of print Jan 22, 2019

J Rehabil Med 2019; 51: 127-135

Correspondence address: Thomas Benz, Research Department, Rehabilitation Clinic "RehaClinic", Quellenstrasse 34, CH-5330 Bad Zurzach, Switzerland. E-mail: t.benz@rehaclinic.ch

Chronic pain occurs in countries with different cethnic, racial, and cultural backgrounds, and has an adult prevalence of 19% in Europe (1). "Ethnic and racial differences" have been described for response to experimental pain, as well as for the perception, expe-

LAY ABSTRACT

This study examined the health of patients with chronic pain who participated in a specific rehabilitation programme for chronic pain. Patients' health was measured before participating in the programme, at the end of the programme, and after the programme. German-speaking patients were compared with Italianspeaking patients. Both groups participated in the same pain management programme with the same therapies, held either in German or in Italian. Italian-speaking and German-speaking patients improved immediately after the programme, but Italian-speaking patient reported lower improvements. The measurements after the programme showed that German-speaking patients could maintain the improvements, whereas the Italian-speaking patients lost the positive effects of the programme over time. In conclusion, Italian-speaking patients may have special needs in pain management and the level of acculturation may influence the results.

rience and treatment of pain (2, 3). Reasons proposed for ethnic differences in the prevalence of widespread pain include psychosocial or cultural differences, genetic predisposition, and the influence of physical or social environment (4, 5).

Interdisciplinary biopsychosocial rehabilitation for chronic low back and fibromyalgia reduces pain intensity, symptom impact, disability and improves functional capacity and work status (6–9). Based on a biopsychosocial treatment approach for chronic pain, interdisciplinary pain management programmes (IPMP) consider that pain, disability, and participation are not only the consequence of a biomedical cause (10, 11). Biological, psychological and social factors influence the development, course and maintenance of chronic pain (12, 13). These dimensions, as well as their interactions, are therefore, recommended to be an integral part of the treatment of low back pain and fibromyalgia (14, 15).

Social factors can be defined as "external facts and circumstances that influence and control an individual's behaviour or attitudes related to pain", e.g. language (16). Language difficulties may lead to misunderstandings relating to treatment intentions and

T. Benz et al.

decisions, advice given after therapy, the healthcare and insurance system, as well as educational treatment. and may not be sufficiently taken into account by the therapists. Social factors were shown to influence functional limitation and disability in low back pain, with social disadvantage being one of the important factors (16).

In Switzerland, the population with foreign citizenship has grown continuously since 1983 and had reached 24.6% of the total population by 2015. This is one of the highest percentages of foreign residents in Europe. In Switzerland, 84.7% of the foreign permanent resident population are of European origin, and the largest group of foreigners are Italian at 15.2% (17). Reactions to health-illness process are different in each culture and are influenced by the patient and by interpretations and attitudes that lead to a certain way of living with the disorder (18). More intercultural competence, awareness and culturally sensitive attitudes are required in medical treatment in order to meet the needs of these different groups better, e.g. by standardized IPMPs held in specific languages.

No studies have investigated the differences in changes as a result of the same IPMP with the same therapy components held in 2 different languages, administered by the same therapists. The objective of this study was to describe and to examine state (at baseline) and short-term (at discharge) as well as mid-term (at 3-, 6-, and 12-month follow-up) changes in biopsychosocial health and quality of life in Italian-speaking patients (ISP) with fibromyalgia, generalized widespread pain and chronic non-specific back pain before and after a standardized 4-week IPMP in the Italian language. Furthermore, the study aimed to compare the results of the ISP with the results of German-speaking patients (GSP) at follow-ups.

METHODS

Patients included in this study were recruited at the pain cen-

tre of the rehabilitation clinic "RehaClinic" in Bad Zurzach,

Switzerland. All patients with chronic non-specific back pain,

fibromyalgia according to the American College of Rheuma-

tology (ACR) definition (19), or generalized widespread pain

(pain in all 4 body quadrants not fulfilling the ACR criteria of

1990 (19)) who attended the "Zurzach Interdisciplinary Pain

Program" (ZISP) were invited to participate. Further inclusion

criteria were age \geq 18 years and chronic pain for \geq 3 months. Ex-

clusion criteria were severe somatic or mental illness/condition

that prevented participation in the IPMP, insufficient language

skills (reading and writing) to complete the assessment tool,

Prior to inclusion in the IPMP and based on the admission report, potential participants were contacted by telephone by

the programme leader (RB) or his assistant (both of whom are

and refusal to participate in the study or the IPMP.

Patients

Journal of Rehabilitation Medicine

bilingual in German and Italian) to complete inclusion and exclusion criteria and to assess oral language skills (listening and speaking). Allocation to the groups was based on oral language skills and not on migration background. For example, all patients participating in the German-speaking programme had excellent oral German language skills, which were an inclusion criteria. Patients with insufficient oral German or Italian language skills (exclusion criteria) were admitted to a different, individual IPMP in our clinic.

Each group underwent a specific Italian- or German-speaking IPMP. This means that all information and instructions were given either in German or Italian. Both language-specific programmes consisted of the same therapy components, the same therapeutic content, and the same number of therapies. All therapies in the programme were provided by the same therapists for the German, as well as for the Italian, group.

Written informed consent was obtained from all participants. The study protocol was approved by the Local Ethics Commission (Health Department in Aarau, Switzerland, EK AG 2008/026).

Intervention

The ZISP is a 4-week, standardized, comprehensive, inpatient IPMP in groups. The IPMP focused on chronic musculoskeletal pain disorders and consisted of active physiotherapy, and aerobic endurance training, Qigong/tai chi exercises, individual psychotherapy including cognitive behavioural therapy, participation in a pain coping group, relaxation therapy, humour therapy, horticultural therapy (since April 2006), information and education about the pathophysiology of pain mechanisms and management of chronic disabling pain, nursing care, and regular medical consultations, including drug therapy. Details of the IPMP are provided elsewhere (6, 7).

Data sampling

In this prospective cohort study, 2 different samples are described and analysed. Sample 1 was collected between 2001 and 2005 with follow-up measurements at 3 and 6 months (Fig. 1), sample 2 between 2006 and 2014 with follow-up measurements at 12 months. In 2006, the follow-up measurements at 3 and 6 months were replaced by one follow-up measurement at 1 year after entry to the programme. This decision was based on the findings for the GSP analysed in sample 1. In this study, only mid-term results were calculated (3 and 6 month follow-up) (6). In the following, long-term results came into our focus of interest, especially because these could not otherwise be found in the literature at that time. A further reason for the restart with the "revised" sample strategy (sample 2) of the IPMP was that a new, improved version (version 2) of the Medical Outcomes Study Short Form 36 Health Survey (SF-36) was available and included in the measurements (20).

For these reasons, the size of sample 1 was limited. The analysis based on sample 1 revealed relatively high effect sizes for many health scales (6). Thus, data sampling for sample 2 was considered sufficient after having reached $n \ge 60$ for both the German- and the Italian-speaking group. Doubling the sample size up to n=120 for each group would narrow the width of the 95% confidence interval of the effect sizes by 0.02. This gain was considered to be too small in face of the burden of doubling the sampling effort.

The difference in number of included patients in sample 1 and the difference in duration of inclusion of the patients in sample 2 were due to the naturalistic study design and the maximal

128



Fig. 1. Flow chart of study participants.

capacity of 6 patients per treatment group in the ZISP. Only one group of patients was treated at a time. Twice a year, the programme was held in Italian for the ISP. This means that 2 groups with ISP and 10 groups with GSP were treated per year. This resulted in an unbalanced number of patients included in sample 1 with the same observation duration (Fig. 1). In order to obtain almost equal numbers of patients, different observation periods were chosen in sample 2.

Measures

Sociodemographic and potentially confounding parameters, such as age, sex, occupation, living conditions, sports, and formal education, were recorded at admission to the clinic on a standardized form used previously in many studies (6). Comorbidities were retrieved from the medical history.

The SF-36 comprehensively measures the dimensions of quality of life, physical, mental and psychosocial health (21). This instrument contains 36 items in 8 health domains: bodily pain, physical functioning, role physical, general health, vitality, social functioning, role emotional, and mental health. It is a commonly used measurement for the self-assessment of health-related quality of life in chronic pain diseases, such as fibromyalgia (22). It has already been used to assess the efficacy of interventions in rheumatology, physiotherapy, drug treatment, tai chi and many others (22). The validated German version was used for the GSP (23). In sample 1, version 1 (21) was used and in sample 2, version 2 (20). For the ISP, the validated Italian version was used (24).

Analysis

JRM

Patients from sample 1 were assessed at baseline (T=0), discharge (T1; short-term), i.e. 4 weeks after entry, 3 months

after entry (T2; mid-term), and 6 months after entry (T3; mid-term). Patients from sample 2 were assessed at baseline (T=0), discharge (T1; short-term), i.e. 4 weeks after entry, and 12 months after entry (T4; mid-term).

SF-36 scores were transformed into scales ranging from 0 ("maximal symptoms or limitation") to 100 ("no symptoms or limitation") to ease comparison of the descriptive data (25). The specific "missing rules" of the instrument had to be fulfilled for determination of the scales. This means that at least 50% of the items had to be completed for each of the SF-36 scales (25).

Sociodemographic and disease-relevant frequency data were compared by the χ^2 test and continuous data by the nonparametric Wilcoxon test. Changes on the SF-36 scales between baseline and follow-up were quantified by multivariate standardized mean differences (SMD) (26). For each SF-36 score, stepwise multivariate linear regression was used to model the individual score changes (baseline to follow-up) as dependent variables. The same independent variables were used for all scales in both samples: group allocation (1=GSP, 0=ISP), baseline score, and sex and education (27). The last 3 variables are well-known as potential confounders for the score changes between baseline and follow-up. The number of confounders is limited by the number of patients in the smallest group/10 (28). The coefficient/slope of the group allocation variable was then equal to the adjusted score difference and was used to calculate the multivariate SMDs (26).

The SMD equals the difference of the mean score changes (baseline to follow-up) between the 2 groups (GSP and ISP) divided by the pooled standard deviation of the score changes (baseline to follow-up) of the 2 compared groups (26). The pooled variance equals the mean of the 2 score change variances, which is weighted by the number of patients. Intervals for 95% confidence (95% CI) for the SMD and *t*-test based type I errors (*p*) for testing SMD>0.00 (zero outside of the 95% CI)

Journal of Rehabilitation Medicine

were calculated (see Tables II and III). An SMD of 0.20-0.49 is considered small, 0.50-0.79 moderate, and ≥ 0.80 large (29). All analyses were performed using the statistical software package IBM SPSS 23.0 for Windows® (SPSS Inc., Chicago, IL, USA).

RESULTS

Patients

The flow chart of participants is shown in Fig. 1. Complete data were available for analysis of 136 GSP and 35 ISP in sample 1, and 63 GSP and 61 ISP in sample 2. Reasons for exclusion with exact numbers are shown in Fig. 1.

Sociodemographic variables and disease-relevant characteristics are given in Table I. All participants were Caucasian. The mean participant's age was comparable in both samples. Comparing sample 1 with sample 2, an important difference was found in sex, with no significant difference in sample 1, but in sample 2 more males participated in the study (p < 0.001). In education, ISP were significantly less educated than GSP, especially in sample 2. In sample 1, higher education was reported

in GSP (73%) vs ISP (9%) in sample 1 and in sample 2 (62% vs 15%). Significant occupational differences were found in sample 1 (p<0.001), with 80% of the ISP being unemployed. In sports, ISP were less active (p=0.034). In both samples, most ISP were living with a partner. In sample 1, 66% of ISP had 3 or more comorbidities in contrast to 43% of the GSP. In both samples, this characteristic was comparable. In summary, ISP were less educated, less occupied, more often married and less active in sports than the GSP, especially in sample 1.

Sensitivity analysis

Age, sex and SF-36 scores for completers were compared with the baseline scores of drop-outs for ISP and GSP in sample 1 and sample 2 each (i.e. 4 comparisons). No significant differences were found in all 4 comparisons in age and sex. In sample 1, significant differences were found in the GSP in Physical functioning (p=0.034) and in the ISP in Role Physical (p=0.032) and in Role emotional (p=0.016). In sample 2, the only significant difference was in the GSP in Role emotional (p=0.011). All other scores differences within the 4 groups were not significant.

Table I. Sociodemographic and disease-related characteristics at baseline

	Sample 1			Sample 2		
	German-speaking patients (n = 136)	Italian-speaking patients (<i>n</i> = 35)	<i>p</i> -value	German-speaking patients $(n=63)$	Italian-speaking patients (n=61)	<i>p</i> -value
Age, years, mean (SD)	46.5 (12.1)	48.9 (7.7)	0.132	48.6 (10.0)	49.9 (9.0)	0.260
Female, %	80.1	74.3	0.448	85.7	45.9	< 0.001
Education, %						
No school	0.0	20.0	< 0.001	3.2	7.3	< 0.001
Basic school (8–9 years)	27.2	71.4		34.9	78.2	
Vocational training	55.1	5.7		46.0	10.9	
College/high school	11.0	2.9		12.7	1.8	
Technical college	2.9	0.0		1.6	0.0	
University	3.7	0.0		1.6	1.8	
Occupation, %						
Not working	36.0	80.0	< 0.001	66.1	67.9	0.949
Part-time work (7-41 h/week)	43.4	14.3		27.5	23.1	
Full-time work (\geq 42 h/week)	16.9	2.9		6.4	9.0	
Living conditions, %						
Alone	22.1	2.9	0.027	27.0	12.5	0.022
With partner	71.3	91.4		54.0	80.4	
With other persons	6.6	5.7		19.0	7.1	
Sport, %						
No sports	48.5	77.4	0.034	50.8	60.0	0.324
<1 h/week	16.9	6.5		23.8	10.9	
1-2 h/week	23.5	12.9		12.7	16.4	
>2 h/ week	11.0	3.2		12.7	12.7	
Diagnosis, %						
Fibromyalgia/Generalized widespread pain	59.6	48.6	0.241	42.9	44.3	0.875
Chronic non-specific back pain	40.4	51.4		57.1	55.7	
Comorbidities, %						
None	9.6	2.9	0.179	7.9	4.9	0.300
1	20.6	11.4		15.9	23.0	
2	27.2	20.0		14.3	31.1	
3	23.5	31.4		23.8	14.8	
4	8.1	20.0		19.0	8.2	
5	5.1	11.4		9.5	1.6	
≥6	5.9	2.9		9.5	1.6	

h/week: hours per week; *p*-value: *p*-values with significance level set at ≤ 0.05 .

Journal of Rehabilitation Medicine

		Baseline (TO)	Difference discharge to baseline (T1-T0)			Difference between 3-month follow- up and baseline (T2–T0)			Difference between 6-month follow-up and baseline (T3–T0)		
Mean (SD)		Mean (SD)	Mean (SD)	SMD [95% CI]	p-value	Mean (SD)	SMD [95% CI]	<i>p</i> -value	Mean (SD)	SMD 95% CI]	p-value
SF-36 Physical	G	41.48 (21.09)	9.68 (18.65)	0.321	0.084	7.38 (21.65)	0.237	0.216	8.13 (21.36)	0.436	0.018
functioning	Ι	35.29 (17.49)	4.71 (15.86)	[-0.050,0.692]		3.48 (17.65)	[-0.142,0.617]		0.00 (19.29)	[0.064,0.809]	
SF-36 Role	G	8.46 (19.30)	11.82 (30.45)	0.572	0.002	12.13 (33.61)	0.413	0.029	8.27 (32.35)	0.404	0.029
physical	Ι	2.14 (9.34)	2.14 (17.54)	[0.197,0.947]		1.52 (15.23)	[0.032,0.794]		0.00 (12.13)	[0.031,0.776]	
SF-36 Bodily	G	19.55 (13.98)	8.75 (16.08)	0.055	0.769	8.92 (18.09)	0.502	0.007	7.49 (20.73)	0.465	0.011
pain	Ι	17.97 (10.48)	10.20 (13.51)	[-0.314,0.425]		3.33 (12.49)	[0.120,0.884]		-1.00 (14.87)	[0.092,0.839]	
SF-36 General	G	41.61 (16.83)	3.00 (14.08)	0.524	0.004	1.36 (15.83)	0.356	0.064	2.86 (17.34)	0.481	0.009
health	Ι	31.00 (10.76)	-0.14 (12.80)	[0.150,0.898]		-2.97 (12.75)	[-0.029,0.742]		-3.57 (13.91)	[0.108,0.855]	
SF-36 Vitality	G	24.81 (16.07)	14.73 (16.84)	0.598	0.001	9.04 (16.12)	0.577	0.002	7.44 (17.98)	0.612	0.001
	Ι	30.94 (15.32)	3.06 (17.47)	[0.222,0.973]		-2.59 (18.87)	[0.189,0.965]		-5.80 (18.09)	[0.237,0.988]	
SF-36 Social	G	43.09 (25.41)	11.85 (27.47)	0.408	0.027	8.97 (25.46)	0.450	0.017	5.67 (28.47)	0.371	0.045
functioning	Ι	31.66 (18.53)	12.54 (18.11)	[0.035,0.780]		4.88 (19.55)	[0.069,0.832]		-1.46 (16.80)	[-0.001,0.743]	
SF-36 Role	G	31.09 (40.99)	13.12 (41.36)	0.423	0.022	12.99 (42.78)	0.197	0.306	8.12 (46.54)	0.175	0.350
emotional	Ι	12.37 (30.35)	9.51 (29.82)	[0.051,0.796]		15.12 (31.27)	[-0.183,0.576]		10.46 (37.76)	[-0.195,0.546]	
SF-36 Mental	G	46.24 (19.83)	9.85 (15.95)	0.323	0.082	5.97 (16.49)	0.623	0.001	3.43 (17.59)	0.465	0.011
health	Ι	47.46 (18.81)	5.51 (14.63)	[-0.048.0.694]		-3.03 (18.72)	[0.234,1.012]		-5.29 (19.19)	[0.092.0.838]	

Table II. Sample 1: SF-36 scores at baseline (T0), at discharge from the pain management programme (T1), at 3-month follow-up (T2) and 6-month follow-up (T3) of German- (n = 136) and Italian-speaking patients (n = 35)

Mean: arithmetic mean; SD: standard deviation; SMD: standardized mean difference; 95% CI: 95% confidence interval; *p*-value: statistical significance comparing German- with Italian-speaking patients; SF-36: Medical Outcomes Study Short Form 36 Health Survey; G: German-speaking patients, I: Italian-speaking patients.

Outcome and comparison of score changes (baseline to follow-up)

At baseline, physical and psychosocial health for the ISP were worse than for the GSP in both samples, with the exception of less vitality and mental health in sample 1 (Table II) and less pain in sample 2 (Table III). However, the differences in the baseline scores were not statistically tested for significance because the baseline scores were included in the multivariate analysis as confounders.

Immediately after the programme at discharge (T1), GSP improved on all scales in both samples. ISP improved on 7/8 scales in sample 1 and 6/8 in sample 2. However, the 3 negative score changes were very small (\geq -0.51 score points). In sample 1, the GSP

improved most on Vitality (mean m=14.73), whereas the ISP improved most on Social functioning (12.54).

In the following course (T2, T3, T4), the improvements decreased slightly for the GSP but more markedly for the ISP with increasing time. For example, in Bodily pain, the score changes were 8.75 (T1)/8.92 (T2)/7.49 (T3) for the GSP, but 10.20/3.33/–1.00 for the ISP in sample 1. In sample 2, the corresponding score changes were 8.27 (T1)/8.22 (T4) for GSP and 7.51/2.25 for ISP. At the 6-month follow-up (T3), 8/8 score changes showed improvement in the GSP, while 5/8 showed worsening, and 2/8 showed no change in the ISP. At the 12-month follow-up (T4), 8/8 score changes showed improvement in the GSP, while 7/8 showed worsening in the ISP.

Table III. Sample 2: SF-36 scores at baseline (T0), at discharge from the pain management programme (T1), and at the 12 month follow-up (T4) of German- (n = 63) and Italian-speaking patients (n = 62)

		Baseline (T0)	Difference fron	n discharge to baseline (Difference of 1 year follow-up to baseline (T4-T0)		
Mean (SD)		Mean (SD)	Mean (SD)	SMD 95% [CI]	p-value	Mean (SD) SMD [95% CI]	p-value
SF-36 Physical functioning	G	43.33 (19.18)	6.43 (17.45)	0.543 [0.186,0.899]	0.002	4.76 (17.52) 0.491 [0.131,0.851] 0.005
	Ι	37.46 (18.97)	1.30 (15.58)			-0.69 (19.41)	
SF-36 Role physical	G	26.95 (15.31)	9.67 (20.13)	0.531 [0.175,0.887]	0.002	5.86 (19.28) 0.558 [0.196,0.919] 0.001
	Ι	17.70 (18.17)	3.77 (26.06)			-1.31 (20.99)	
SF-36 Bodily pain	G	16.81 (12.99)	8.27 (15.59)	-0.030 [-0.379,0.320]	0.868	8.22 (20.90) 0.263 [-0.093,0.61	9] 0.141
	Ι	18.33 (12.46)	7.51 (9.45)			2.25 (14.67)	
SF-36 General health	G	39.38 (17.66)	3.84 (18.43)	0.712 [0.351,1.073]	0.000	0.38 (18.89) 0.441 [0.082,0.800] 0.012
	Ι	30.25 (14.13)	-0.08 (15.67)			-2.97 (15.24)	
SF-36 Vitality	G	26.78 (14.70)	10.75 (17.98)	0.429 [0.076,0.783]	0.014	4.73 (19.13) 0.479 [0.120,0.839] 0.007
	Ι	24.36 (17.12)	5.82 (17.19)			-0.19 (20.27)	
SF-36 Social functioning	G	40.29 (26.33)	10.06 (22.91)	0.239 [-0.112,0.590]	0.177	3.59 (32.56) 0.231 [-0.125,0.58	6] 0.198
	Ι	37.49 (24.86)	5.31 (23.98)			-3.48 (25.00)	
SF-36 Role emotional	G	45.10 (30.90)	4.25 (27.43)	0.473 [0.118,0.827]	0.007	2.78 (32.70) 0.473 [0.114,0.833] 0.007
	Ι	29.89 (30.55)	-0.51 (31.33)			-6.16 (39.26)	
SF-36 Mental health	G	46.35 (22.17)	6.83 (14.65)	0.230 [-0.121,0.581]	0.193	4.44 (20.93) 0.575 [0.213,0.937] 0.001
		39.26 (20.39)	4.18 (19.58)			-3.62 (23.15)	

Mean: arithmetic mean; SD: standard deviation; SMD: standardized mean difference; CI: 95% confidence interval; *p*-value: statistical significance comparing German- with Italian-speaking patients; SF-36: Medical Outcomes Study Short Form 36 Health Survey; G: German-speaking patients; I: Italian-speaking patients.

This led to increasing levels of SMDs during the course, since they quantify the differences of the score changes between GSP and ISP. However, adjustment for the confounders attenuated those differences during the course. For example, on Bodily pain, the observed score changes (listed above) led to bivariate, unadjusted SMDs of -0.093 (T1)/0.324 (T2)/0.429 (T3), but 0.055/0.502/0.465 multivariate, adjusted SMDs in sample 1. The differences on Bodily pain in sample 2 were not statistically significant; the adjusted SMD at baseline was -0.030 (p=0.868) and increased to a SMD of 0.263 (p=0.141). The same was true for Social functioning.

Significant change differences between GSP and ISP in sample 1 at each follow-up measurement were seen on the following scales: Role physical (adjusted SMD=0.572, 0.413, 0.404 for T1 to T3), Vitality (0.598, 0.577, 0.612), and Social functioning (0.408, 0.450, 0.371). In sample 2, Physical functioning (adjusted SMD=0.543, 0.491 for T1 and T4), Role physical (0.531, 0.558), General Health (0.712, 0.441), Vitality (0.429, 0.479), and Role emotional (0.473, 0.473) showed significant differences.

In total, 5 of 8 SMDs were statistically significant $(p \le 0.027)$ in both samples at discharge and at the 3-month follow-up. At the 6-month follow-up in sample 1, 7 SMDs $(p \le 0.045)$ and at 1-year follow-up, 6 SMDs $(p \le 0.012)$ were statistically significant.

Of the total of 8 health dimensions, 5 showed significant differences at discharge, 5 at the 3-month follow-up, and 7 at the 6-month follow-up in sample 1. In sample 2, the corresponding numbers were 5 (discharge) and 6 (12-month follow-up).

DISCUSSION

This study compared short- and mid-term changes in the biopsychosocial health and quality of life of ISP with GSP with chronic pain before and after a standardized IPMP. GSP improved in all measured scales at discharge and the effects remained almost stable in the follow-up measurements. In contrast, ISP showed less improvement on most scales at discharge and lost these positive effects completely over time.

These score change differences resulted in statistically significant adjusted SMDs on Role physical and Vitality, as well as, although somewhat less on General health over the course. At the mid-term follow-ups (6 and 12 months), all but one of the scales showed significant differences in favour of the GSP. These prominent differences cannot be explained by differences in therapy because both groups underwent the same structured standardized IPMP held in the specific language.

The main focus of the study was to explore possible differences between GSP and ISP. The naturalistic design of the study is one of the factors that led to baseline differences. The design of the study did not allow us to draw causal conclusions that explain the differences in the score changes of the 2 groups, the SMDs. Some cofactors may be found in the sociodemographic and disease-related characteristics (Table I), and some in parameters that were not assessed (e.g. sickness benefits from the insurance). Some can be found in existing literature: ethnicity including cultural, behavioural and attitudinal norms and systems of meaning (5, 30, 31) and socioeconomic level (4, 5) including educational level and work status. For the 3 most relevant confounders, sex, education and baseline score, analysis of the differences was adjusted by multivariate regression analysis. All 3 cofactors are well-known to affect and confound the outcome by epidemiological reasons and showed the biggest differences between the 2 groups at baseline. Even after correction of the unequal distributions of those 3 cofactors, substantial and statistically significant differences in outcome were observed between the 2 groups.

Although language has been described as "a proxy for acculturation" and as "a significant marker of cultural identity", empirical data to support this are lacking (32). An improving knowledge of the German language in combination with an increasing length of stay in a German-speaking region is assumed to enhance the level of acculturation (33). This means that attitudes, values, customs, beliefs, and (health) behaviours are adapted to another culture and the influence of the origin culture diminishes (33–35). It can be assumed that migrants who have acquired high levels of language skills are also well acculturated in other cultural dimensions. Therefore, cultural differences within the German-speaking group of patients are thought to be small and of minor importance for health status and healthcare utilization, independent of migration status.

It has been suggested that disparities in pain among racial and ethnic minorities may be influenced not only by the patient, but also by the healthcare provider and the healthcare system (2). In Switzerland, healthcare insurance is mandatory and all patients have equal access to the healthcare system. Nevertheless, lower levels of acculturation and no German language knowledge influenced the accessibility and use of the healthcare system (36).

A comparable study examined Italian migrant workers (n=36) and Swiss GSP (n=49) with chronic low back pain (37). Both groups underwent an identical 3-month outpatient treatment programme in different languages. The levels of state and change in physical

functioning were similar to those of our study. The Italian group showed no improvements in pain, disability, mental state, flexibility and strength after 3 months of treatment, while the GSP reported highly significant improvements in all measurements.

Consistent findings revealed 2 cross-sectional studies: Italian patients with chronic low back pain reported the highest levels of emotional impairment, second-highest levels of social impairment, and thirdhighest levels of physical impairment compared with different culture groups (38). Migrants from different countries, who lived in Switzerland for longer had worse health than indigenous Swiss people of the same age (36).

In our data, important improvements in various health dimensions were observed at discharge after intensive inpatient treatment of GSP and ISP. Improvements in ISP were lower and in some health dimensions even negative (worsening). Most of the short-term improvements were maintained up to midterm (6 or 12 months) in GSP. In contrast, ISP lost almost all of these improvements. This loss might be caused by some of the complex interactions mentioned above. Chronic pain can be initiated and increased by psychological distress caused by perceived discrimination (39). One possible interpretation is that the special needs of patients with a migration background may partially be met within an inpatient IPMP held in their native tongue, but are not generally met at home after discharge. At the end of the programme all patients receive individually tailored recommendations for subsequent outpatient management irrespective of the language.

The differences in outcome between GSP and ISP in this study cannot be explained by differences in the language-specific programmes, since all therapies and therapist were the same for both groups. Differences in effects cannot be attributed to therapeutic characteristics or comprehensibility. Inequalities in baseline characteristics, which were (Table I) or were not assessed in the study, are superficial. In addition, heterogeneities of characteristics within the fibromyalgia syndrome and back pain may have an impact. Knowing these factors, higher sample sizes would be needed to adjust for them.

Reducing language barriers by administering the treatment in Italian language seems not to be sufficient, because the improvements observed in the Italian group were smaller than those of the German group. In addition to language barriers, cultural barriers have been shown to have a negative impact on the recovery process (35): "Cultural differences may result in diverging and conflicting representations of health, illness and therapy, and this may hinder the

healing process or even cause its failure". Four mediating factors in intercultural care have been identified to facilitate or hinder the care relationship and, by that, the rehabilitation process: (*i*) humanity in care, (*ii*) communication, (*iii*) the role of the family, and (*iv*) the hospital's organizational culture (40). Adapting the therapy content in combination with specific intercultural competence training of all involved medical personal might enhance the treatment effect in the IPMP by implementing a comprehensive "cultural sensitive care in which patients are cared for in a holistic and dignity-enhancing way" (40).

For detailed insight into these complex clinical situations, various factors should be further investigated: (*i*) influence of migration details (e.g. migration background, migration trajectories), (*ii*) level of acculturation (language skills among other factors), (*iii*) identification of key aspects of intercultural competence and communication in different care settings and health professions, (*iv*) adaptation of methods and content of treatments, (*v*) definition of health and pain, particular needs and specific expectations from a patient's perspective, and (*vi*) choice of assessments to measure change of main problems of these patients.

This study has several weaknesses. There was a lack of detailed information about socioeconomic status and migration information, including land of origin, place of birth and reasons for migration. Socioeconomic status was approximated by education and occupation level. The high number of patients lost to follow-up is a potential threat to the internal and external validity of this study. However, sensitivity analysis showed rare differences in baseline characteristics between the subjects who completed the study and those who dropped out during observation time. This means, that selection bias due to sex, age and SF-36 baseline score differences was small. A further weakness is the lack of knowledge about the treatment during follow-up periods after discharge. Continuation of the recommended therapies after dismissal was not assessed. Although both versions of the SF-36 have been derived from and validated to the English original, cross-validation between the German and Italian version has not been performed. Psychometric differences between the 2 versions may exist and contribute to the differences of the outcome measurement.

A strength of this study is the naturalistic, prospective study design with comparison of 2 groups participating in the same IPMP with the same therapy components in different languages. ISP received the same therapies as the GSP in their own language. A further strength is the consistency of the differences across 2 different samples. Although the observational, non-randomized study design without a control group

134 *T. Benz et al.*

did not permit a causal statement on the effectiveness of the treatment, quantification and comparison of changes in health were possible. The differences in changes in health between the 2 groups were adjusted by cofactors that were unequally distributed at baseline and which possibly confounded the outcome changes.

In conclusion, ISP reported lower improvements in state of health after an IPMP than GSP. In the subsequent follow-ups, these differences were further accentuated. The reasons for these differences remain unclear, but may have consequences for the future management of ISP in IPMPs. Patients with a migration background may have special needs in therapeutic management and addressing them might enhance the positive outcome in the short- and mid-term.

ACKNOWLEDGEMENTS

The authors thank all patients for participating in the study and the members of the Pain Center of the rehabilitation clinic "RehaClinic" in Bad Zurzach, Switzerland. We would also thank Joy Buchanan and Stacy Müller for English editing. The authors received no financial support for the research, authorship, and/ or publication of this article.

The authors have no conflicts of interest to declare.

REFERENCES

- Breivik H, Collett B, Ventafridda V, Cohen R, Gallacher D. Survey of chronic pain in Europe: Prevalence, impact on daily life, and treatment. Eur J Pain 2006; 10: 287–333.
- Green CR, Anderson KO, Baker TA, Campbell LC, Decker S, Fillingim RB, et al. The unequal burden of pain: confronting racial and ethnic disparities in pain. Pain Med 2003; 4: 277–294.
- Sleptsova M, Woessmer B, Grossman P, Langewitz W. Culturally sensitive group therapy for Turkish patients suffering from chronic pain: a randomised controlled intervention trial. Swiss Med Wkly 2013; 143: w13875.
- Palmer B, Macfarlane G, Afzal C, Esmail A, Silman A, Lunt M. Acculturation and the prevalence of pain amongst South Asian minority ethnic groups in the UK. Rheumatology 2007; 46: 1009–1014.
- Allison TR, Symmons DPM, Brammah T, Haynes P, Rogers A, Roxby M, et al. Musculoskeletal pain is more generalised among people from ethnic minorities than among white people in Greater Manchester. Ann Rheum Dis 2002; 61: 151–156.
- Angst F, Brioschi R, Main CJ, Lehmann S, Aeschlimann A. Interdisciplinary rehabilitation in fibromyalgia and chronic back pain: a prospective outcome study. J Pain 2006; 7: 807–815.
- Angst F, Verra M, Lehmann S, Brioschi R, Aeschlimann A. Clinical effectiveness of an interdisciplinary pain management programme compared with standard inpatient rehabilitation in chronic pain: a naturalistic, prospective controlled cohort study. J Rehabil Med 2009; 41: 569–575.
- Kamper SJ, Apeldoorn AT, Chiarotto A, Smeets RJEM, Ostelo RWJG, Guzman J, et al. Multidisciplinary biopsychosocial rehabilitation for chronic low back pain. Cochrane Database Syst Rev 2014; 9: CD000963.
- Lera S, Gelman SM, López MJ, Abenoza M, Zorrilla JG, Castro-Fornieles J, et al. Multidisciplinary treatment of fibromyalgia: does cognitive behavior therapy increase

the response to treatment? J Psychosom Res 2009; 67: 433-441.

- 10. Engel GL. The need for a new medical model: a challenge for biomedicine. Science 1977; 196: 129–136.
- 11. Waddell G. The Back Pain Revolution. 2nd edn. Edinburgh: Churchill Livingstone; 2004, p. 480.
- 12. Linton SJ, Shaw WS. Impact of psychological factors in the experience of pain. Phys Ther 2011; 91: 700–711.
- Nicholas MK, Linton SJ, Watson PJ, Main CJ, 'Decade of the Flags' Working Group. Early identification and management of psychological risk factors ('yellow flags') in patients with low back pain: a reappraisal. Phys Ther 2011; 91: 737–753.
- Waddell G, Burton AK. Concepts of rehabilitation for the management of low back pain. Best Pract Res Clin Rheumatol 2005; 19: 655–670.
- 15. Burckhardt CS. Multidisciplinary approaches for management of fibromyalgia. Curr Pharm Des 2006; 12: 59–66.
- Shaw WS, Campbell P, Nelson CC, Main CJ, Linton SJ. Effects of workplace, family and cultural influences on low back pain: what opportunities exist to address social factors in general consultations? Best Pract Res Clin Rheumatol 2013; 27: 637–648.
- Foreign population. Federal Statistical Office. 2017 [cited 2017 Mar 14]. Available from: https://www.bfs.admin.ch/ bfs/en/home/statistics/population/migration-intergration/ foreign.html.
- Plaza Del Pino FJ, Soriano E, Higginbottom GM. Sociocultural and linguistic boundaries influencing intercultural communication between nurses and Moroccan patients in southern Spain: a focused ethnography. BMC Nurs 2013; 12: 14.
- Wolfe F, Smythe HA, Yunus MB, Bennett RM, Bombardier C, Goldenberg DL, et al. The American College of Rheumatology 1990 Criteria for the Classification of Fibromyalgia. Report of the Multicenter Criteria Committee. Arthritis Rheum 1990; 33: 160–172.
- Ware JE, Kosinski M, Dewey JE. How to Score Version 2 of the SF-36 Health Survey. 2nd edn. Lincoln, RI: Quality Metric Incorporated; 2000.
- Ware JE, Snow KK, Kosinski M, Gandek B, New England Medical Center Hospital, Health Institute. SF-36 health survey: manual and interpretation guide. 3rd ed. Boston: Health Institute, New England Medical Center; 2004.
- 22. Busija L, Pausenberger E, Haines TP, Haymes S, Buchbinder R, Osborne RH. Adult measures of general health and health-related quality of life: Medical Outcomes Study Short Form 36-Item (SF-36) and Short Form 12-Item (SF-12) Health Surveys, Nottingham Health Profile (NHP), Sickness Impact Profile (SIP), Medical Outcomes Study Short Form 6D (SF-6D), Health Utilities Index Mark 3 (HUI3), Quality of Well-Being Scale (QWB), and Assessment of Quality of Life (AQoL). Arthritis Care Res 2011; 63 Suppl 11: S383–S412.
- Bullinger M, Kirchberger I. Fragebogen zum Gesundheitszustand: SF-36; Handanweisung Hogrefe, Verlag für Psychologie; 1998.
- 24. Apolone G, Mosconi P. The Italian SF-36 Health Survey: translation, validation and norming. J Clin Epidemiol 1998; 51: 1025–1036.
- 25. Angst F, Pap G, Mannion AF, Herren DB, Aeschlimann A, Schwyzer H-K, et al. Comprehensive assessment of clinical outcome and quality of life after total shoulder arthroplasty: usefulness and validity of subjective outcome measures. Arthritis Rheum 2004; 51: 819–828.
- Borenstein M. Effect sizes for continuous data. In: Cooper H, Hedges LV, Valentine JC, eds. The handbook of research synthesis and meta-analysis. 2nd edn. New York: Russell Sage Foundation; 2009; p. 222–236.
- 27. Angst F, Aeschlimann A, Angst J. The minimal clinically important difference raised the significance of outcome effects above the statistical level, with methodological implications for future studies. J Clin Epidemiol 2017;

Journal of Rehabilitation Medicine

82: 128-136.

- Hosmer DW, Lemeshow S, Sturdivant RX. Applied logistic regression. 3rd edn. Hoboken, NJ: Wiley; 2013, p. 528.
- Cohen J. Statistical power analysis for the behavioral sciences. 2nd edn. Statistical power analysis for the behavioral sciences. New York: Lawrence Erlbaum Associates; 1988.
- Rahim-Williams FB, Riley JL, Herrera D, Campbell CM, Hastie BA, Fillingim RB. Ethnic identity predicts experimental pain sensitivity in African Americans and Hispanics. Pain 2007; 129: 177–184.
- 31. Lasch KE. Culture, pain, and culturally sensitive pain care. Pain Manag Nurs 2000; 1 Suppl 1: S16–S22.
- Tarnutzer S, Bopp M, SNC Study Group. Healthy migrants but unhealthy offspring? A retrospective cohort study among Italians in Switzerland. BMC Public Health 2012; 12: 1104.
- Abraído-Lanza AF, Armbrister AN, Flórez KR, Aguirre AN. Toward a theory-driven model of acculturation in public health research. Am J Public Health 2006; 96: 1342–1346.
- Sundquist K, Li X. Coronary heart disease risks in firstand second-generation immigrants in Sweden: a follow-up study. J Intern Med 2006; 259: 418–427.
- Côté D. Intercultural communication in health care: challenges and solutions in work rehabilitation practices and

training: a comprehensive review. Disabil Rehabil 2013; 35: 153–163.

- Guggisberg J, Gardiol L, Graf I, Oesch T, Künzi K, Volken T, et al. Gesundheitsmonitoring der Migrationsbevölkerung (GMM) in der Schweiz: Schlussbericht. 2011 [cited 2017 Mar 2]; Available from: http://www.ssoar.info/ssoar/ handle/document/36016.
- 37. Kopp H, Kippele C, Brioschi R, Oliveri M, Läubli T, Christen S, et al. Psychosoziale Faktoren in der Chronifizierung von Rückenschmerzen bei Schweizern und italienischen Immigranten (Projekt-Nr. 4026-27071 im NFP 26B). In: Schlussbericht des Nationalen Forschungsprogramms Nr 26B. Basel: Eular Verlag; 1996, p. 114–121.
- Sanders SH, Brena SF, Spier CJ, Beltrutti D, McConnell H, Quintero O. Chronic low back pain patients around the world: cross-cultural similarities and differences. Clin J Pain 1992; 8: 317–323.
- Brown TT, Partanen J, Chuong L, Villaverde V, Chantal Griffin A, Mendelson A. Discrimination hurts: The effect of discrimination on the development of chronic pain. Soc Sci Med 2018; 204: 1–8.
- 40. Degrie L, Gastmans C, Mahieu L, Dierckx de Casterlé B, Denier Y. How do ethnic minority patients experience the intercultural care encounter in hospitals? a systematic review of qualitative research. BMC Med Ethics 2017; 18: 2.